

Optimizing Lake Surface Water Temperature Simulations Over Large Lakes in China With FLake Model

Ling Huang, Xuhui Wang, Yuxing Sang, Shuchang Tang, Lei Jin, Hui Yang, Catherine Ottlé, Anthony Bernus, Shenglei Wang, Chenzhi Wang, et al.

▶ To cite this version:

Ling Huang, Xuhui Wang, Yuxing Sang, Shuchang Tang, Lei Jin, et al.. Optimizing Lake Surface Water Temperature Simulations Over Large Lakes in China With FLake Model. Earth and Space Science, 2021, 8 (8), 10.1029/2021ea001737. hal-03323554

HAL Id: hal-03323554 https://hal.science/hal-03323554

Submitted on 21 Aug 2021 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License



Earth and Space Science

RESEARCH ARTICLE

10.1029/2021EA001737

Key Points:

- Freshwater lake model (FLake) has different capabilities in reproducing the thermodynamics of seasonally ice-covered lakes and non-icecovered lakes
- The performance of FLake in simulating the lake surface water temperature of 94 large lakes across China was optimized during 1996–2011
- FLake successfully reproduced variations in lake surface water temperature after adjusting the albedo, lake mean depth and light extinction coefficient

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

X. Wang, xuhui.wang@pku.edu.cn

Citation:

Huang, L., Wang, X., Sang, Y., Tang, S., Jin, L., Yang, H., et al. (2021). Optimizing lake surface water temperature simulations over large lakes in China with FLake model. *Earth and Space Science*, *8*, e2021EA001737. https://doi.org/10.1029/2021EA001737

Received 26 APR 2021 Accepted 25 JUL 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Optimizing Lake Surface Water Temperature Simulations Over Large Lakes in China With FLake Model

Ling Huang¹, Xuhui Wang¹, Yuxing Sang¹, Shuchang Tang¹, Lei Jin¹, Hui Yang¹, Catherine Ottlé², Anthony Bernus², Shenglei Wang³, Chenzhi Wang¹, and Yuan Zhang¹

¹Sino-French Institute for Earth System Science, College of Urban and Environmental Sciences, Peking University, Beijing, China, ²Laboratoire des Sciences du Climat et de l'Environnement, IPSL, CNRS-CEA-UVSQ, Gif-sur-Yvette, France, ³Key Laboratory of Digital Earth Science, Aerospace Information Research Institute, Chinese Academy of Sciences, Beijing, China

Abstract Lake surface water temperature (LSWT) is sensitive to climate change; however, simulated LSWT and its response to climate change remain uncertain. In this study, FLake, a one-dimensional freshwater lake model, is optimized to simulate the LSWTs of 94 large lakes with surface areas greater than 100 km² in China. While most of these lakes are seasonally ice-covered over the Tibetan Plateau, FLake with default parameters significantly underestimated LSWT in spring and winter and slightly overestimated LSWT in summer and autumn in seasonally ice-covered lakes. We performed sensitivity experiments and calibration in the trial lake (Qinghai Lake). Then, parameter calibrations of three lakes specific properties (albedo, lake mean depth and light extinction coefficient) were performed in all the studied lakes. The optimized FLake substantially improved the simulations of seasonal and interannual variations in LSWT. The root mean square error decreased from $3.64 \pm 1.54^{\circ}$ C to $1.97 \pm 0.72^{\circ}$ C, and the mean bias of 96% of the lakes decreased to less than 1°C. Our study showed that the optimized FLake can reproduce the temporal variations in LSWT across China with optimized parameters, providing the possibility to simulate and project the response of LSWT to rapid climate change.

Plain Language Summary Lake surface water temperature (LSWT), one of the key physical properties of lake water, directly or indirectly impacts physical and biological processes within lakes. LSWT also controls the exchanges of water and heat at the lake-air interface. Therefore, it is necessary to accurately simulate and predict changes in LSWT. This study evaluated the simulation capabilities of the freshwater lake model (FLake) for LSWTs in 94 large lakes across China. FLake has better LSWT simulation skills for non-ice-covered lakes than for seasonally ice-covered lakes. Through a series of experiments in Qinghai Lake, three dominant parameters for model simulation were identified, and the observation-based calibration greatly improved the performance of the FLake model. This study demonstrates the potential of FLake in simulating and predicting LSWT changes and provides a reference for future lake simulations in China.

1. Introduction

Lakes, as important natural resources, hold enormous amounts of liquid surface freshwater (Gleick, 1993), support global biodiversity and provide ecosystem services to humans (Xu et al., 2018). However, lakes are facing numerous threats due to climate change (Woolway et al., 2020). Evidences have emerged that lake temperature (O'Reilly et al., 2015; Verburg & Hecky, 2009), lake area (Yang & Lu, 2014; Zhang et al., 2017), lake water level (Chen et al., 2017; Lei et al., 2019), lake mixing regimes (Butcher et al., 2015; Woolway et al., 2017) and lake ice phenology (Cai et al., 2019; Sharma et al., 2019) are responsive to climate change. Among these phenomena, the most direct impact of climate change on lakes is changes in lake surface water temperature (LSWT).

Lake surface warming affects the internal processes of lake ecosystems by changing the physical and biogeochemical properties of lakes, such as more stable stratification, decreased lake mixing, increased algal growth, more methane emissions and decreased primary productivity (Bastviken et al., 2004; Carter & Schindler, 2012; O'Reilly et al., 2003; Paerl & Paul, 2012), which may affect lake ecosystem services (Jeppesen et al., 2014). Moreover, increases in LSWT will affect the hydrological process in a basin (Chen et al., 2017; Gronewold & Stow, 2014) and even affect the regional climate, leading to an increasing intensity of lake-effect precipitation (Steiner et al., 2013).

There are ~2,600 lakes (>1 km²) in China, accounting for ~1% of the total land area (Ma et al., 2011). The largest lake (Qinghai Lake) has a surface area of 4,267 km². The altitudes of lakes span a large range from 0 to 5,198 m. Seventy-five percent of lakes are seasonally ice-covered and concentrated in high-latitude or high-altitude regions (Ma et al., 2011). Millions of people live around lakes and depend on the ecosystem services of lakes. From the perspective of both lake ecosystem services and regional climate effects, it is necessary to analyze and predict the temporal and spatial variations in LSWT. Although the diversity and importance of Chinese lakes make them ideal regions to study LSWT, previous studies were mostly on a single lake or over parts of China (Su et al., 2019; Wu et al., 2019; Zhang et al., 2014), rendering the LSWT dynamics and its response to climate change uncertain.

The lake model is the most common and effective tool to study LSWT and lake-air interactions. During recent decades, research institutions have developed many one-dimensional lake models and three-dimensional lake models based on different key physical processes and parameterization schemes. Although three-dimensional lake models contain more comprehensive and complicated physical processes within lakes, both their computational cost and time efficiency constrain their application. Moreover, accurate parameters of three-dimensional lake models are difficult to obtain. Considering these limitations, one-dimensional lake models have been more widely used in previous studies (Lv et al., 2019; Turuncoglu et al., 2013; Wu et al., 2020). One-dimensional lake models assume zero lateral heat/water exchange. Studies have proven that one-dimensional lake models can meet the needs of a range of lake simulations (Fenocchi et al., 2018; Thiery et al., 2015). These common one-dimensional lake model include the freshwater lake model (FLake) based on self-similarity theory, the eddy-diffusive Hostetler model (Hostetler et al., 1993; Subin et al., 2012), the Mixed-Layer model (Goyette et al., 2000), and the $k - \varepsilon$ turbulence closure model (Perroud et al., 2009). The Lake Model Intercomparison Project (LakeMIP) compared the performances of different lake models and revealed that no single model outperformed the others globally, with performances varying among lakes due to differences in lake parameters (Stepanenko et al., 2010, 2013).

FLake is a model capable of reproducing the surface thermal behavior and vertical temperature structure in lakes of various depths (Mironov, 2008; Mironov et al., 2010). In recent decades, many studies have evaluated the performances of FLake in reproducing lake thermodynamics (Lv et al., 2019; Rooney & Bornemann, 2013; Thiery et al., 2015). Despite the difference between simulations from FLake and observations (Bernhardt et al., 2012; Layden et al., 2016; Pour et al., 2012), FLake is thought to be a reliable model for global-scale studies after tuning trials (Layden et al., 2016; Woolway & Merchant, 2019).

In this study, we improved the performance of FLake in large lakes over China to accurately simulate the thermal behaviors of lake surfaces. First, we conducted preliminary trial work in Qinghai Lake, the largest lake in China, and set up a series of sensitivity experiments to calibrate key lake-specific parameters with an improved observation-driven method integrating both satellite observations and the model-tuning algorithm recommended by Layden et al. (2016). Based on the trial in Qinghai Lake, we further optimized FLake model performances in all 94 large lakes during 1996–2011. We organized the paper as follows: Section 2 describes the studied lakes, data sets, an overview of the FLake model, model modifications, and methods for model optimization. Section 3 compares and improves the model performance against satellite-derived data sets. Section 4 summarizes and discusses the results.

2. Materials and Methods

2.1. Study Area

Ninety-four large lakes (>100 km²) with LSWT observations of the Along Track Scanning Radiometer Reprocessing for Climate: Lake Surface Water Temperature and Ice Cover (ARC-Lake) data set in China were chosen as study areas to calibrate and improve the FLake model performance. Lake boundaries were extracted from the HydroLAKES data set (Messager et al., 2016), as well as lake areas, lake depths and altitudes. These lakes account for nearly 50% of lake areas in China and vary in surface area (100.8–4,266.6 km²), depth (1–120 m), altitude (0–5,198 m), and geographic location (Figure 1). The studied lakes were divided into six subregions to demonstrate the simulations of LSWT according to geographical and climatic features (Ma



Earth and Space Science



Figure 1. Spatial distribution of the studied lakes. The lakes are divided into six subregions: Tibetan Plateau (TP), Xinjiang (XJ), Inner-Mongolia Plateau (MP), Northeast Plain and Mountain (NE), Eastern Plain (EP), and Yunnan-Guizhou Plateau (YG). The underlying background displays the elevation from Shuttle Radar Topography Mission Digital Elevation Database v4.1.

et al., 2011; Zhang et al., 2019). These regions included the Tibetan Plateau (TP), Xinjiang (XJ), Inner-Mongolian Plateau (MP), Northeast Plain and Mountain (NE), Eastern Plain (EP), and Yunnan-Guizhou Plateau (YG). The lake numbers in the subregions are 58, 9, 3, 1, 20, and 3. Non-ice-covered lakes are mostly located in the EP region and YG region. Seasonally ice-covered lakes are concentrated in the other four subregions. The general characteristics of the studied lakes are shown in Table S1.

2.2. Data

2.2.1. ARC-Lake Surface Water Temperature

In this study, the ARC-Lake data set (version 3.0) retrieved from the ERS and Envisat satellites during 1996–2011 was used as a reference (MacCallum & Merchant, 2012) to evaluate and validate the FLake model. The LSWTs are estimated only in water pixels with clear-sky conditions. When the lake was close to freezing or frozen, the temperature values were replaced by 0°C. The data set contains daytime and nighttime observations on a 0.05° grid and on a lake-by-lake basis. Many studies have shown that ARC-Lake observations compare well with in situ observation records, with a small bias (<1°C) (MacCallum & Merchant, 2012). The ARC-Lake data set has been widely used to validate lake simulations (Thiery et al., 2015; Verseghy & MacKay, 2017). We used the all-day mean LSWT, calculated as the average of daytime and nighttime observations. The lake mean LSWT with bimonthly temporal resolutions was used to optimize the model parameters in our study. Daily reconstructed ARC-Lake observations were used to demonstrate the simulation performance of the ice-free season.

2.2.2. Geomorphometric Characteristics of Lakes

HydroLAKES data set version 1.0 estimates the volumes of all global lakes with surface areas greater than 10 ha using a geostatistical model based on surrounding topography information (Messager et al., 2016).



The data set also provides other additional lake characteristics required by FLake, including lake boundary, lake mean depth, lake surface area and latitude/longitude of lake center coordinates. A basic assumption of the geostatistical model is that the topography around the lake is a reliable predictor variable of its bathymetry. The model has been trained and validated on 12,150 existing records of lake depth obtained from 14 national and international datasets. Overall, the lake depths from HydroLAKES match well with reference depths (Messager et al., 2016).

2.2.3. Moderate Resolution Imaging Spectroradiometer (MODIS) Albedo Products

We used Moderate Resolution Imaging Spectroradiometer (MODIS) BRDF/albedo products (MCD43B3, version 5) during the period of 2001–2005 as criteria to correct the albedo parameterization scheme in FLake. The products combine the albedo retrievals from the Aqua and Terra satellites and provide albedo values for different bands, which agree with the in situ observations (He et al., 2014; Schaaf et al., 2002). In this study, we used the white-sky and black-sky albedos for shortwave broadband with a spatial resolution of 1 km and an interval of 8 days. We carefully selected the pixels with quality flags between 0 and 2 to ensure relatively high data quality. The diffuse skylight ratio was calculated using the monthly diffuse and direct downward shortwave radiation products from the National Center for Environmental Forecasting and was introduced to calculate blue-sky albedo. The detailed method is presented in Li et al. (2016).

2.2.4. MODIS Water Clarity Product

Wang et al. (2020) developed a robust water clarity algorithm based on two water color parameters retrieved from the MODIS surface reflectance product MOD09. The light extinction coefficient (K_d) is closely correlated with water clarity. We obtained the K_d from the water clarity product published by Wang et al. (2020). Maps of Secchi desk depth (Z_{SD}) for 153 large lakes across China during the period of 2000–2017 were generated for the first time. The Z_{SD} product showed good consistency with in situ measurements, demonstrating reliability for use. The multiyear average Z_{SDS} of 84 lakes are available for our study. Following the empirical equations recommended by Layden et al. (2016), Z_{SD} was converted to K_d by using $K_d = (0.757/Z_{SD}) + 0.07$ m⁻¹ when Z_{SD} ranged from 2 to 10 m. $K_d = 1.7/Z_{SD}$ was applied for lakes outside the former range. For the 10 lakes without water clarity values, K_d was set to a constant value of 3 m⁻¹.

2.2.5. China Meteorological Forcing Data Set (CMFD)

The FLake forcing variables include near-surface air temperature, specific humidity, wind speed, surface pressure, downward shortwave radiation and downward longwave radiation. We obtained 3-h forcing variables during the period of 1996–2011 from the China Meteorological Forcing Data set (CMFD). The CMFD is a high spatial resolution (0.1°) forcing data set developed by the Institute of Tibetan Plateau Research, Chinese Academy of Sciences (He et al., 2020; Yang et al., 2010) and is the amalgamation of a variety of data sources, including satellite products, reanalysis data sets and meteorological station measurements. This data set has good accuracy for surface meteorological variables and has been widely used in land surface process and hydrological modeling in China (Huang et al., 2017; Lazhu et al., 2016). The forcing data were extracted by the boundaries of lakes and then calculated through spatial averaging. However, considering that this data set are produced by merging various records of land stations, wind speed data are not suitable for representing the actual wind speed over lakes. Scaling of wind speed equation (Equation 1) recommended by Hsu (1988) was adopted to correct the bias.

$$U_{\rm lake} = 1.62 + U_{\rm land} \times 1.17,$$
 (1)

Here, U_{lake} is the wind speed (m s⁻¹) over the lake, and U_{land} (m s⁻¹) is the wind speed over land.

2.3. Model Description

FLake is a one-dimensional freshwater lake model consisting of two vertical water layers (Mironov, 2008; Mironov et al., 2010). FLake divides the water column into an upper mixed layer and an underlying thermocline layer. The upper layer is well mixed and vertically uniform. The vertical temperature profile of the thermocline layer, extending down to the bottom, is parameterized based on the self-similarity theory. Moreover, FLake also describes the thermal structure of one ice layer, one snow layer on lake ice, and two thermally active layers for lake sediments. These layers are also parameterized by the same self-similarity



theory. A prognostic entrainment equation for convective conditions and a relaxation-type equation for stable conditions are used to calculate the mixed-layer depth (Mironov, 2008). The temperature profile in the thermocline layer is computed through the self-similarity theory, that is,

$$\frac{\theta_s(t) - \theta(z, t)}{\Delta \theta(t)} = \Phi_\theta(\zeta) \quad h(t) \le z \le h(t) + \Delta h(t), \tag{2}$$

Here, *t* is time (s). *z* is depth (m). $\theta_s(t)$ is the water temperature in mixed layer (K). h(t) is the depth of mixed layer (m). $\Delta\theta(t) = \theta_s(t) - \theta_b(t)$ is the temperature difference (K) across the thermocline with the depth of $\Delta h(t)$, and $\theta_b(t)$ is the water temperature (K) at the bottom of the thermocline layer. Φ_θ is a dimensionless "universal" shape function of the relative (dimensionless) depth $\boldsymbol{\zeta} = [z - h(t)] / \Delta h(t)$, and Φ_θ satisfies two boundary conditions: $\Phi_\theta(0) = 0$ and $\Phi_\theta(1) = 1$.

The dimensionless shape function $\Phi_{ heta}$ is obtained by

$$\Phi_{\theta}(\zeta) = \left(\frac{40}{3}C_{\theta} - \frac{20}{3}\right)\zeta + (18 - 30C_{\theta})\zeta^{2} + (20C_{\theta} - 12)\zeta^{3} + \left(\frac{5}{3} - \frac{10}{3}C_{\theta}\right)\zeta^{4},\tag{3}$$

Here, C_{θ} is the shape factor and is assumed to vary with time, that is,

$$\frac{dC_{\theta}}{dt} = \operatorname{sign}\left(\frac{dh(t)}{dt}\right) \frac{C_{\theta}^{\max} - C_{\theta}^{\min}}{t_{\operatorname{rc}}} \quad C_{\theta}^{\min} \le C_{\theta} \le C_{\theta}^{\max},\tag{4}$$

In Equation 4, t_{rc} is the relaxation time (s), which is parameterized as the time of the thermocline shape curve varying from one to the other. $C_{\theta}^{\max}(C_{\theta}^{\min})$ is an empirical constant and represents the maximum (minimum) value of shape factor C_{θ} .

The water albedo is set to a constant value (0.07) in FLake, while ice albedo is assumed to be an empirical formulation of lake ice surface temperature:

$$\alpha_{\rm ice} = \alpha_{\rm max} - \left(\alpha_{\rm max} - \alpha_{\rm min}\right) \left[\exp\left(-\frac{95.6(T_f - T_s)}{T_f}\right) \right],\tag{5}$$

In Equation 5, α_{max} is the albedo of white ice, set to 0.6. α_{min} is the albedo of blue ice, set to 0.1. T_f is the freezing temperature (K), and T_s is the ice surface temperature (K). The snow albedo is set equal to the ice albedo. The Beer-Lambert decay law is used to describe the vertical heat transfers in water and ice/snow. The fluxes of momentum and sensible and latent heat fluxes at the lake surface are estimated by a parameterization scheme, which comprehensively considers the relationship between roughness length, potential temperature, specific humidity and wind velocity.

Owing to the simple but realistic representation of major physics processes in lakes, FLake has a high computational efficiency compared with other lake models and has been extensively shown to have good accuracy in reproducing lake surface temperature changes in lakes of different morphometries and climate conditions (Balsamo et al., 2012; Huang et al., 2019; Lazhu et al., 2016; Lv et al., 2019). However, FLake has a limited ability to simulate the thermodynamic processes of deep lakes with depths greater than 60 m due to the absence of monimolimnion layer representation (Balsamo et al., 2012). However, this fact should have minor impacts in our study, as only two lakes in our study have depths greater than 60 m.

2.4. Lake-Specific Parameters

Because of the simplicity of the parameterization scheme and the few external lake-specific parameters used, it is relatively easy to tune FLake to specific applications. These external lake-specific parameters are albedo, lake mean depth, K_d , fetch, relaxation coefficient (c_relax_C), depth of the thermally active layer of bottom sediments (depth_bs), and temperature at the outer edge of the thermally active layer of bottom sediments (T_bs) (Layden et al., 2016; Mironov, 2008; Mironov et al., 2010). We evaluated and modified the former three parameters by referring to the model-tuning algorithm of Layden et al. (2016) to improve the model's capabilities in LSWT simulations during the period of 1996–2011. Fetch and c_relax_C were set to the recommended values from Layden et al. (2016). The bottom sediment module was switched off in the

Table 1

Experimental Design of Freshwater Lake Model (FLake)							
Experiments	Ice albedo	Water albedo	Depth (m)	$K_d (\mathrm{m}^{-1})$			
CTRL	Default	0.07	10	3			
Exp_albedo	Modified	MCD43B3	10	3			
Exp_depth	Modified	MCD43B3	HydroLAKES	3			
Exp_Kd	Modified	MCD43B3	HydroLAKES	f(water clarity) ^a			
Exp_tuning	Modified	MCD43B3	HydroLAKES*factor ₁ factor ₁ :0.25–4.0	f(water clarity) ^a *factor ₂ factor ₂ :0.25–4.0			
Exp_optimization	Modified	MCD43B3	HydroLAKES* effective factor ₁	f(water clarity) ^a *effective factor ₂			

Note. K_d , light extinction coefficient.

^aThe empirical formulations recommended by Layden et al. (2016) were used to calculate K_d based on a satellite-derived water clarity product (Wang et al., 2020).

simulations. Namely, the heat flux from the bottom sediment was set to zero flux, and the values of depth_bs and T bs had no effect on the simulations. Detailed modifications of the parameters are presented in Sections 2.4.1-2.4.3, and the sensitivity experiments are shown in Table 1. Additionally, to obtain physically reasonable initial conditions, we repeatedly ran the model with a cycled one-year atmosphere forcing of 1996 until an equilibrium state was obtained. Then, the initial conditions for 1997 were determined using the values of the prognostic variables at the end of the final year.

Layden et al. (2016) used satellite-observed LSWT from ARC-Lake as the criterion to tune key lake properties (albedo, lake mean depth, and K_d) to make FLake reproduce simulations as closely as possible to the observed variations in LSWT. This model-tuning algorithm avoided the need for a detailed description of lake characteristic information, which is usually not available in many lakes. However, an increasing amount of lake characteristic information can now be obtained from satellite products, such as albedo, lake depth, and water clarity. We improved the tuning process by acquiring these parameters, which were more efficient and physically reasonable for representation of lake properties. Similarly, some necessary adjustments for lake mean depth and K_d were adopted in our experiments to achieve the best performance in simulating LSWT.

2.4.1. Albedo

As mentioned above, the ice albedo in FLake is calculated based on an empirical function of lake surface ice temperature. The scheme originates from a sea model (Mironov, 2008) and can describe seasonal variations in ice albedo to some extent. In China, most seasonally ice-covered lakes are in the TP region. However, some studies pointed out that FLake seriously overestimated the ice albedo in TP lakes (Lang et al., 2018; Li et al., 2018). For example, Lang et al. (2018) compared the simulated albedo against field measurements in Ngoring Lake in 2017 and found that the maximum albedo bias of FLake reached 0.48. The overestimated ice albedo indicates that more solar radiation is reflected off the lake, which affects the melting of lake ice and results in a delay in ice-off date. In sensitivity experiments, we set a series of combinations of white ice albedo (0.2-0.8) and blue ice albedo (0.1-0.6) to tune these two empirical constants until the modeled ice albedo closely approximated the MODIS albedo values. For water albedo, the default values of water albedos in the model were replaced by the averaged MODIS albedos during the open-water period.

2.4.2. Lake Mean Depth

Lake depth directly represents the thermal capacity of the lake and controls how long it takes to cool the lake in autumn and winter, thus controlling the ice-on date. FLake recommends using mean depth as an input parameter. In field observations, it is difficult to directly obtain the mean depth of lakes, especially for large lakes. We obtained lake mean depths from the HydroLAKES data set (Messager et al., 2016) and used them in sensitivity experiments. We also estimated the effective lake depth (multiplying the mean depth by a scale factor) to closely reproduce the observed LSWT dynamics by minimizing the root mean square error (RMSE) between the simulations and observations. The scale factor ranges from 0.25 to 4.0 in the tuning process. The effective lake depth is often not the same as the measured lake mean depth. Lake depth is not the only factor that affects the heat exchange between the lake surface and atmosphere. Other factors include lake bottom sediment, topography, altitude, and lake area. The effective lake depth will compensate



for the influences of these factors on the surface heat exchange. Additionally, FLake does not consider the monimolimnion for deep lakes; thus, an artificial depth (60 m) is adopted for deep lake simulation (Le Moigne et al., 2016; Perroud et al., 2009).

2.4.3. Light Extinction Coefficient

The absorption and attenuation of solar radiation within water bodies are the main processes controlling water surface temperature. K_d was used to represent the amount of attenuation of solar radiation within the water column. Generally, a smaller K_d means that a greater amount of surface heat goes into deep water and results in a lower surface temperature of the lake. Conversely, a higher K_d leads to a higher surface water temperature. Existing field observations of K_d are scarce. Some constant values of K_d (for example, 3 m⁻¹) are usually used to drive the model. In the default configurations, K_d was set to 3 m⁻¹. The water transparency of lakes in China has large spatial heterogeneity, subject to the intensity of human activity and phytoplankton density (Feng et al., 2019). The recently published water clarity product (Wang et al., 2020) was used to estimate the K_ds of lakes in China. Similar to tuning the lake mean depth, K_d is multiplied by a series of scale factors (0.25–4.0) in the tuning process to obtain the effective K_d for each lake. The model was tuned using the optimal combination of depth and K_d factors, and 100 possible combinations for each lake were used in the tuning process.

2.4.4. Metrics Used for Parametrization

A model-tuning algorithm was used to estimate white/blue ice albedo, effective depth and K_d to make the simulations as close to the observations as possible, namely, reaching a minimum RMSE between the modeled and observed albedo or LSWT. The tuning process is widely used in lake model simulations (Layden et al., 2016; Li et al., 2019; Woolway & Merchant, 2019) and can further improve the LSWT simulation.

Three other statistics, including the mean bias error (BIAS), Pearson correlation (R), and Taylor score (TS), were also used to evaluate model skill. The formulas to calculate these statistics are shown below:

$$\mathbf{RMSE} = \sqrt{\frac{1}{N} \sum_{i=1}^{N} (S_i - O_i)^2},$$
(6)

$$\mathbf{BIAS} = \frac{1}{N} \sum_{i=1}^{N} \left(S_i - O_i \right),\tag{7}$$

$$\boldsymbol{R} = \frac{\sum_{i=1}^{N} (S_{i} - \overline{S}) (O_{i} - \overline{O})}{\sqrt{\sum_{i=1}^{N} (S_{i} - \overline{S})^{2}} \sqrt{\sum_{i=1}^{N} (O_{i} - \overline{O})^{2}}},$$
(8)

$$\mathbf{TS} = \frac{4(1+R)}{\left(\sigma + (1/\sigma)\right)^2 (1+R_0)},\tag{9}$$

$$\sigma = \frac{\sqrt{\frac{1}{N}\sum_{i=1}^{N} \left(S_i - \overline{S}\right)^2}}{\sqrt{\frac{1}{N}\sum_{i=1}^{N} \left(O_i - \overline{O}\right)^2}},$$
(10)

where *N* is the number of discrete points (in time or space). S_i and O_i are the simulation and observation at point *i*, respectively. $\overline{S}(\overline{O})$ is the average simulation (observation) over N sample points. σ is the standard

Table 2

The Average Pearson Coefficient (R), Root Mean Square Error (RMSE), Mean Bias Error (BIAS), and Taylor Score (TS) of Lake Surface Water Temperature (LSWT, °C) Between Simulations From the Control Experiment (CTRL) and Observations From ARC-Lake of Lakes in Six Subregions During the Open-Water Period From 1996 to 2011

Subregion	R	RMSE	BIAS	TS
ТР	0.90 ± 0.07	4.17 ± 1.39	-1.60 ± 1.12	0.74 ± 0.13
XJ	0.89 ± 0.08	4.51 ± 1.42	-1.44 ± 0.75	0.80 ± 0.14
MP	0.94 ± 0.01	4.03 ± 0.25	-1.25 ± 0.39	0.93 ± 0.03
NE	0.94	4.11	-1.16	0.94
EP	0.98 ± 0.00	1.91 ± 0.25	0.20 ± 0.50	0.99 ± 0.00
YG	0.93 ± 0.06	1.74 ± 0.45	-0.27 ± 0.66	0.93 ± 0.07

Note. \pm represents 1 standard deviation.

deviation of the simulation divided by that of the observation. R_0 is the achievable maximum correlation coefficient and is set to 1. Both lower RMSE and lower BIAS represent a better performance of the simulation. R shows the temporal/spatial variation similarity between the simulation and observation. TS can simultaneously assess models in amplitude and temporal/spatial patterns (Taylor, 2001). A higher TS implies better model performance to match the observation (Huang et al., 2016).

Considering that ice temperatures were excluded in the ARC-Lake product, model evaluations were performed during the ice-free season. Layden et al. (2015) showed that the first day with lake mean LSWT above/below 1°C was an effective indicator of ice-off/on date for large lakes. The lake ice phenology defined by Layden et al. (2015) has good consistency with in situ observations of ice-on and ice-off days for 21 Eurasian and North American lakes and has been widely adopted in research (Woolway & Merchant, 2019). We followed the method of Layden et al. (2015) and used 1°C as the threshold to define the ice-on/off date and ice-free season.

3. Results

3.1. Evaluation of FLake Performance in the Control Experiment

In the control experiment (CTRL), ice albedo was calculated based on the default ice parameterization in FLake. Water albedo was set to a reference value of 0.07. Lake mean depth was set to 10 m. K_d was set to 3 m⁻¹. Since the purpose of the ARC-Lake data set is to monitor the surface water temperature of the lake, excluding the ice temperature, the evaluations were only carried out during the ice-free season.

Table 2 lists four statistical coefficients to demonstrate the average performances of FLake in simulating the LSWT in six subregions using CTRL settings. Overall, FLake roughly reproduced the seasonal cycle of LSWT, with R ranging from 0.89 to 0.98, and generally underestimated LSWT in the majority of lakes (Figure S1). However, the performances of CTRL showed large differences among the six subregions. CTRL clearly underestimated the average monthly LSWT of lakes in the TP region, XJ region, MP region, and NE region, slightly underestimated that of lakes in the YG region and slightly overestimated that of lakes in the EP region. CTRL has the worst representation for lakes in the TP region, but it has reasonable representation for lakes in the EP and YG regions. These results indicate that the LSWT simulation of FLake over seasonally ice-covered lakes is poorer than that over non-ice-covered lakes. The RMSEs of non-ice-covered lakes ranged from 1.44°C to 2.27°C, while the RMSEs of seasonally ice-covered lakes ranged from 1.72°C to 8.13°C (Figure S1). To further clarify the reasons for this difference, we examined the monthly RMSE and BIAS between the CTRL simulations and observations in six subregions to determine the months that were most responsible for the discrepancy (Figure 2). We found that the seasonally ice-covered lakes usually had the largest RMSEs and BIASs in winter and spring, reflecting the biases of the ice-off/on date and duration of ice cover (Figures S2 and S3). The seasonally ice-covered lakes in the TP and XJ regions showed the same bias pattern, with a positive bias in summer and negative biases in other seasons (Figure 2). For these lakes, CTRL underestimated the LSWT by a maximal BIAS in winter, and BIAS gradually decreased in spring and turned to positive values in June. The overestimations could last three to four months and then change to underestimations again in October. For lakes in the MP and NE regions, the overestimations continued even longer until November. The simulation performances of non-ice-covered lakes were quite different from those of seasonally ice-covered lakes. The RMSEs of lakes in the EP and YG regions did not show obvious seasonal variations, with a range of 0.97°C-2.53°C. The variations in BIASs for non-ice-covered lakes were largely random and lake specific, suggesting that parameters other than ice albedo may be more responsible for the model biases in lakes over the EP and YG regions.





Figure 2. The average monthly root mean square error (RMSE) and mean bias error (BIAS) between the simulated lake surface water temperature in the control experiment (CTRL) and observations from ARC-Lake of lakes in six subregions during the open-water period from 1996 to 2011. The bars represent the RMSE, and the red lines represent the BIAS. The black lines are used to distinguish the positive and negative BIASs. Evaluations over the ice-covered period are omitted.

3.2. Trial Lake Simulated With the Improved Lake Scheme

3.2.1. Control Experiment for the Trial Lake

Qinghai Lake was used as a trial lake to exhibit the performance of CTRL in simulating LSWT. CTRL captured the seasonal variations in LSWT well (Figure 3). However, CTRL significantly overestimated the ice-covered period and underestimated the open-water period. The modeled ice-off date was ~28 days later than the observed ice-off date, while the modeled ice-on date was ~24 days earlier than the observed ice-on date. Additionally, the simulated LSWTs were higher than the observations in summer, while the simulated LSWTs were generally lower than the observations in other seasons (Table 3). This result was consistent with the simulations of Su et al. (2019). The simulated LSWT had a positive BIAS of 1.89°C in summer, a negative BIAS of -2.70° C in spring, and a negative BIAS of -4.01° C in autumn and early winter (Table S2). CTRL underestimated the annual average LSWT during the open-water period with a BIAS of -1.90° C and an RMSE of 1.97° C. Moreover, CTRL showed similar performances in simulating daytime and nighttime LSWT (Figures S4 and S5; Table S3).

3.2.2. Modification of Albedo

The large negative biases between the modeled and observed LSWTs in spring and winter were caused by the biases in simulating ice phenology. Ice albedo is thought to be the main lake parameter that affects the ice-off date. During the period 2001–2005, the average ice albedo was 0.51 in FLake, much higher than the 0.21 in the MODIS albedo products (Figure S6). The overestimation of ice albedo can attenuate absorbed solar radiation and eventually result in postponement of the ice-off date. Hence, the ice albedo parameterization scheme was first modified in our study.

We used the MODIS albedo products to tune the empirical constants within the albedo parameterization scheme. Smaller albedo constants were determined, 0.3 for white ice albedo and 0.1 for blue ice albedo. The average water albedo observed (0.017) during the open-water period replaced the default water albedo (0.07)





Figure 3. The multiyear mean time series of lake surface water temperature (LSWT) from the ARC-Lake observations and the simulations in the control experiment (CTRL) for Qinghai Lake during the period of 1996–2011. The shaded area and error bar represent one standard deviation. ARC-Lake observations below 1°C represent the ice-covered period of Qinghai Lake. Inserted plots are the comparisons of ice-off (ice-on) dates between CTRL simulations and observations. The first day of LSWT > 1°C in spring/summer indicates the ice-off date. The first day of LSWT < 1°C in winter indicates the ice-on date. Error bars represent one standard deviation of ice-off/on dates.

in the albedo experiment (Exp_albedo). Although large deviations in albedo still existed in some months, the modified albedo was generally close to the observed ice albedo (Figure S6). Exp_albedo clearly displayed improved skill in simulating LSWT in spring (Figure S7). The underestimation of spring LSWT and delay in ice-off date were both significantly reduced. The BIAS and RMSE between the simulated and observed LSWT in spring decreased from -2.70°C and 2.84°C to 0.25°C and 1.33°C, respectively (Table 3). Moreover,

Table 3

The Monthly Root Mean Square Error (RMSE) and Mean Bias Error (BIAS) of Lake Surface Water Temperature (LSWT, °C) Between the Simulations From Five Experiments and Observations From ARC-Lake for Qinghai Lake During the Open-Water Period From 1996 to 2011

	RMSE				BIAS					
Month	CTRL	Exp_ albedo	Exp_ depth	Exp_ Kd	Exp_ optimization	CTRL	Exp_ albedo	Exp_ depth	Exp_Kd	Exp_ optimization
Apr	3.93	2.31	2.07	1.98	1.77	-3.87	-2.11	-1.93	-1.81	-1.55
May	2.09	2.56	1.31	0.81	1.22	-1.75	2.37	0.98	0.11	-0.79
Jun	2.70	3.63	2.91	1.34	1.01	2.54	3.55	2.78	1.17	0.18
Jul	2.12	2.57	2.20	1.19	0.78	2.05	2.49	2.09	1.11	0.38
Aug	1.25	1.52	1.73	1.22	0.88	0.96	1.36	1.57	1.08	0.68
Sep	1.20	0.85	0.73	0.79	1.00	-1.00	-0.6	0.10	0.42	0.67
Oct	2.59	2.08	1.58	0.77	0.52	-2.55	-2.01	-1.49	-0.40	0.21
Nov	3.08	2.49	1.57	0.59	0.78	-3.05	-2.43	-1.43	-0.17	0.60
Dec	9.95	7.56	5.12	4.12	0.77	-9.89	-7.51	-4.99	-3.82	-0.14

Note. Evaluations over the ice-covered period (January-March) are omitted.

the overestimation in summer LSWT increased in Exp_albedo, with the BIAS increasing from 1.89°C to 2.50°C. Since the observed water albedo was lower than the default value, more solar radiation heated the surface water. Similarly, due to the decreasing water albedo, the negative BIAS in autumn and early winter was reduced (Table S4).

3.2.3. Modification of Lake Mean Depth

Reproducing the ice-on dates as realistically as possible is another difficulty in simulating lake thermodynamics. Lake depth plays a determinant role in controlling the ice-on dates. Lakes usually begin to freeze in shallows along the shore and then expand toward the center. Shallow waters have lower thermal inertia, requiring less time to lose heat to warm air and reach thermal equilibrium with the air, thus forming lake ice earlier. In the depth experiment (Exp_depth), the mean depth of Qinghai Lake was set to a HydroLAKES depth of 16.8 m. Figure S8 shows that the simulated dates of ice-on occurred later and were closer to the dates of ARC-Lake. The BIAS in autumn and early winter significantly decreased to -1.89° C. The differences between the modeled and observed LSWTs were also reduced in other seasons (Table S5). The RMSE (BIAS) of spring LSWT was 1.10° C (-0.34° C). The overestimation of summer LSWT was slightly mitigated (Table 3). The deeper the lake is, the more solar radiation is needed for the water column to fully mix in spring. Under the same radiation conditions, the surface water temperature of deep lakes will be lower than that of shallow lakes, resulting in a reduced spring LSWT in Qinghai Lake. The decrease in summer LSWT may be caused by deepening of the mixed layer in Exp_depth. The RMSE (BIAS) of the annual average LSWT covering the open-water period is 0.56° C (-0.32° C). Overall, the simulated LSWT was closer to the observed LSWT when the lake mean depth was adjusted from 10 to 16.8 m.

3.2.4. Modification of the Light Extinction Coefficient

Solar radiation is the fundamental energy source of the whole lake system. K_d is a key parameter that determines the ability of solar radiation to penetrate water bodies, thus affecting the vertical thermal structure of lakes. In general, the more transparent the lake is, the easier it is for solar radiation to penetrate deeper into the water, resulting in a lower LSWT. Qinghai Lake is a typical dimictic lake with two turnovers in spring and autumn and one stratification in summer. The lake mixes completely from the surface to the bottom during the turnover period. Thus, the effect of K_d on LSWT is more notable in the stratification period. In CTRL, K_d was set to a constant of 3 m⁻¹. Qinghai Lake is in the TP region, with fewer aquatic plants and pollution discharges. The default value of K_d was too high for Qinghai Lake. K_d in the Exp_Kd experiment was tuned to a lower value of 0.31 m⁻¹, which was derived from a water clarity product (Wang et al., 2020). Exp_Kd produced a significantly decreased summer LSWT (Figure S9). BIAS (RMSE) in summer was decreased to 1.13°C (1.21°C). Moreover, by reducing K_d , RMSE decreased in both spring and autumn (Table S6). The higher lake transparency led to more rapid warming in spring, a deeper mixed layer and a warmer thermocline in summer, which made the stratification more stable and ultimately led to slower fall cooling. The RMSE and BIAS for annual LSWT during the open-water period were 0.60°C and -0.27°C, respectively.

3.2.5. Joint Optimization of the Parameters

After gradually adopting satellite-derived albedo, lake mean depth and K_d , the simulations became increasingly closer to the observations (Figure 4). Although Exp_Kd greatly improved the simulation performance, it still overestimated summer LSWT and estimated ice-on dates too early. Tuning lake mean depth and K_d can theoretically optimize the simulations. The underlying mechanism of depth optimization is to compensate for the influences of other factors on heat/water exchange at the lake-air interface. K_d was estimated from a Z_{SD} product based on two empirical formulas, with large uncertainties. Tuning K_d aims to reduce the errors and uncertainties in the empirical formulas. We conducted a joint tuning process to adjust the lake mean depth and K_d in the tuning experiment (Exp_tuning). The depth factor and K_d factor both ranged from 0.25 to 4.0. We designed 100 combinations of depth factors and K_d factors. The factors were determined based on the minimum RMSE between the simulated and observed LSWT during 1996–2011.

For Qinghai Lake, the effective depth factor is 1.5, and the K_d factor is 0.75. The experiment using optimal effective factors is called the optimization experiment (Exp_optimization, Figure S10). RMSE decreased by more than 57% compared with CTRL (Figure 5; Table S7). The declines in RMSE were most obvious from October to December, reaching 82%, while the declines were less obvious in August and September, reaching 23%. The most pronounced LSWT improvement in spring was in Exp_albedo and that in summer was





Figure 4. The multiyear mean time series of lake surface water temperature from the ARC-Lake observations and simulations in five experiments for Qinghai Lake during the period of 1996–2011. The shaded area and error bar represent one standard deviation. ARC-Lake observations below 1°C represent the ice-covered period of Qinghai Lake.

in Exp_Kd. Exp_albedo, Exp_depth and Exp_Kd showed similar improvements in the LSWT simulations in autumn and early winter. Additionally, as the time series of 16 years was relatively short, *R* and TS in tuning experiments were prone to be affected by extreme values or null values, which produced large variability in monthly *R* and TS (Figure S11). The BIAS of ice phenology was reduced to 9.6 days for ice-off dates and 0.2 days for ice-on dates (Figure S12). Figure 6 shows the overall performances of standard deviation, *R* and RMSE compared with observations at annual scales using the Taylor diagram. The results showed that these experiments had obvious differences in simulating LSWT. Exp_optimization yielded the best simulation ability and outperformed other experiments for the whole open-water period. This result implied that the optimized FLake can realistically reproduce the LSWT in Qinghai Lake both seasonally and interannually.

3.3. Application of the Improved Lake Scheme in All Lakes

After adjusting these three parameters, FLake achieved good simulations in Qinghai Lake. Following the same strategy, we optimized FLake performances over all 94 large lakes across China. The ice albedo, water albedo, effective lake depth factor and K_d factor, which were finally applied in Exp_optimization for all lakes, are listed in Table S1. Table 4 shows the average RMSEs of subregions under five experiments.

Figure 7 exhibits the spatial patterns of RMSE, BIAS, *R* and TS in Exp_optimization. The results showed that the RMSEs of 70 lakes were less than 2°C, accounting for 74% of the studied lakes. Ninety lakes with BIAS values within $\pm 1^{\circ}$ C accounted for 96% of the studied lakes. Moreover, almost all lakes had *R* and TS values above 0.85. Compared with the CTRL, the improvements were more significant in seasonally ice-covered lakes. For seasonally ice-covered lakes, the BIAS decreased from $-1.56 \pm 1.05^{\circ}$ C to $0.03 \pm 0.62^{\circ}$ C, and the RMSE decreased from $4.21 \pm 1.35^{\circ}$ C to $2.07 \pm 0.70^{\circ}$ C. For non-ice-covered lakes, BIAS and RMSE decreased slightly from $0.14 \pm 0.54^{\circ}$ C and $1.89 \pm 0.27^{\circ}$ C to $-0.08 \pm 0.34^{\circ}$ C and $1.52 \pm 0.29^{\circ}$ C, respectively. Overall, the optimization increased the *R* and TS values nationally, particularly in the TP region. Exp_optimization also greatly improved the simulation performance of ice phenology. The *R*-square of the simulated and observed ice-on dates reached 0.74, and the *R*-square of the simulated and observed ice-off dates was 0.61 (Figure S13). Figure 8 shows the interannual variation in the regional average LSWT during the open-water





Figure 5. Histograms of root mean square error (RMSE) and mean bias error (BIAS) of lake surface water temperature between the simulations from five experiments and the observations from ARC-Lake for Qinghai Lake during 1996–2011.

period from 1996 to 2011. The results showed that the averaged LSWTs of the subregions between Exp_optimization simulations and ARC-Lake observations were highly consistent. In summary, Exp_optimization can reconstruct the dynamic changes in surface water temperature well in large lakes across China.

The largest improvements in Exp_optimization were in spring and winter for seasonally ice-covered lakes (Figure 9). The dominant parameters of LSWT vary with climate, season and lake type (Figure S14). Albedo dominated the LSWT simulation for seasonally ice-covered lakes in spring and winter. Non-ice-covered lakes were more sensitive to lake depth and K_d . In the albedo experiment, the RMSE in the TP region decreased by 76% in January, and the RMSE in the XJ region decreased by 52% in December. The LSWT in the MP region was very sensitive to lake depth in spring. In the depth experiment, the RMSE of LSWT in the MP region decreased by 43%–57% in spring. In Exp_depth and Exp_Kd, the RMSE in the YG region decreased by 18% and 16%, respectively. Notably, the RMSE of the NE region slightly increased in November after tuning the lake depth. This result occurred in Chagan Lake, the only large lake in ARC-Lake for northeastern China. The few effective observations of Chagan Lake in winter made the RMSE assessment less reliable. Taylor diagrams comprehensively show the simulation performances of five experiments in six subregions (Figure 10). Overall, Exp_optimization performed best and can realistically represent the LSWT temporal



Figure 6. Taylor diagram of monthly lake surface water temperature of Qinghai Lake in the open-water period for five experiments.

variations in all subregions. In addition, the improvements in FLake were significant in the TP region, XJ region and NE region.

4. Discussion

Using the ARC-Lake satellite observations of lakes with surface areas larger than 100 km² from 1996 to 2011, the one-dimensional freshwater lake model FLake was evaluated and calibrated to simulate the LS-WTs in 94 large lakes across China. Based on satellite observations, we tuned three lake-specific parameters (albedo, lake mean depth and K_d) in FLake to improve the ability to represent seasonal and interannual variations in LSWT. Compared with a previous study (Layden et al., 2016), our observation-driven tuning strategy improved the efficiency of the tuning process and reduced the risk of equifinality. The tuning

Table 4

The Average Root Mean Square Error (RMSE) Between the Monthly Simulated Lake Surface Water Temperature (LSWT, °C) From Five Experiments and Monthly Observed LSWT From ARC-Lake of Lakes in Six Subregions During the Open-Water Period From 1996 to 2011

	RMSE						
Subregion	CTRL	Exp_albedo	Exp_depth	Exp_Kd	Exp_optimization		
ТР	4.17 ± 1.39	3.18 ± 1.20	2.57 ± 0.85	2.46 ± 0.80	1.91 ± 0.62		
XJ	4.51 ± 1.42	4.01 ± 1.66	3.71 ± 1.13	3.54 ± 0.81	3.02 ± 0.44		
MP	4.03 ± 0.25	3.71 ± 0.51	3.33 ± 1.12	3.33 ± 1.12	2.04 ± 0.38		
NE	4.11	3.62	3.11	3.11	3.07		
EP	1.91 ± 0.25	1.91 ± 0.25	1.77 ± 0.25	1.78 ± 0.26	1.59 ± 0.23		
YG	1.74 ± 0.45	1.73 ± 0.42	1.29 ± 0.12	1.29 ± 0.11	1.05 ± 0.17		
Note. \pm represents 1 standard deviation.							

HUANG ET AL.





Figure 7. Spatial patterns of root mean square error (RMSE), mean bias error (BIAS), Pearson coefficient (*R*) and Taylor score (TS) of lake surface water temperature between the simulations from the optimization experiment (Exp_optimization) and the observations from ARC-Lake during the open-water period from 1996 to 2011. The inset in map shows the frequency distribution of the statistic.

process demonstrated the potential of FLake to reproduce realistic LSWT simulations across China, and FLake can be considered an appropriate tool for investigating and predicting LSWT changes in large lakes across China.

The correct lake-specific parameters greatly improved the simulation performance of FLake. We found that these parameters were consistent with observations or previous calibration studies. For example, the effective lake depths of Qinghai Lake and Nam Co Lake are 25 and 44 m in the study, respectively, which were very close to the previous results of 21 (Su et al., 2019) and 40 m (Lazhu et al., 2016). The K_d values of Poyang Lake and Liangzi Lake were 0.91 and 0.75 m⁻¹, respectively, which were similar to the results of 0.8 and 1.1 m⁻¹ by Li et al. (2019).

Regional differences were apparent in the ice albedo, lake mean depth and K_d of large lakes in China (Table S1). For seasonally ice-covered lakes, ice albedos were lower in the TP region and higher in other regions. In addition, lakes in eastern China were shallow and low transparent, while lakes in western China were deep and high transparent. White ice albedo represents the largest ice albedo in seasonally ice-covered lakes. In tuning the white/blue ice albedo, the white ice albedos were generally not higher than 0.3 in the TP region, while the white ice albedos in other regions were generally between 0.4 and 0.6. This finding implied that the default ice albedo in FLake seemed too high for the majority of large lakes in China. In the tuning process of effective lake depth, we found that lakes with tuning to a shallower effective depth were dominant, accounting for ~62% of the total lakes. Additionally, 28% of lakes tended toward a deeper effective depth. Seventy-seven percent of the lakes with deeper tuning were less than 20 m deep (Figure 11). The bottom sediment module was switched off in our study, and the deeper effective depth can compensate for the influences of bottom sediment on surface temperature. A deeper effective depth has the same effect on the heat storage capacity of lakes as the heat retention in the sediment (Layden et al., 2016). In western China, the K_d values were mostly between 1.5 and 4.0 m⁻¹. Most K_ds in eastern China were lower than 1.5 m⁻¹. In general, the effective K_d factor decreased with increasing K_d in both western lakes and eastern lakes (Figure 11). Across all studied lakes, 72% were tuned to be more transparent. Twenty-five percent of





Figure 8. Time series of the yearly average lake surface water temperature during the open-water period from the optimization experiment (Exp_optimization) and from ARC-Lake in subregions from 1996 to 2011. Red and black lines represent simulations and observations, respectively. The red shade and gray shade represent one standard deviation of simulations and observations, respectively.

lakes were tuned to be more turbid. Lakes with tuning that was more transparent were widely distributed in China. Lakes with more turbid tuning were distributed in some lakes with high transparency values $(K_d < 0.25 \text{ m}^{-1})$ in the southern TP region, MP region and YG region (Table S1). The optimal lake-specific characteristics provide guidance for the improvement of LSWT simulations in China in the future.

Although the FLake model has been shown to be an effective model for simulating lake surface temperature, we recognize that some uncertainties and limitations still exist in our research. First, accurate simulation representation strongly depends on the reliability of the meteorological forcing data. The forcing data used in the study are an assimilation product from land meteorological observations. There are some biases between meteorological variables near lakes and those near land. For example, due to the small surface roughness of the water surface, the wind speeds over lakes are significantly higher than the wind speeds over surrounding land (Lazhu et al., 2016; Su et al., 2019). The underestimation of the 10 m wind speed can lead to a decreasing upward heat loss and therefore an overestimation of LSWT. The relationship between lake wind and land wind varies with lake and season (Phillips & Irbe, 1978; Quinn, 1979). No one wind speed adjustment is most suitable for all lakes. Layden et al. (2016) compared three kinds of wind speed adjustments in 35 globally distributed large lakes and found that Hsu's (1988) equation was most applicable to seasonally ice-covered lakes. Since most large lakes in China are seasonally ice-covered, we applied Hsu's (1988) equation to correct these wind speeds. The equation does introduce some biases and uncertainties in our study. More lake wind observations are needed in the future.

In addition, most lakes in the TP region are saline lakes (Song et al., 2013). The salinity of lake water strongly affects the maximum water density. Huang et al. (2019) found that adjusting the temperature of maximum density in Nam Co Lake can significantly enhance the convective mixing induced by the vertical density gradient and improve the simulation of the vertical thermal structure during cold seasons. Su et al. (2019) suggested that salinity affected the timing of overturn and duration of lake stratification in Qinghai Lake. These results implied that salinity indirectly affected the variation in surface temperature by affecting the vertical temperature structure of the lake. Moreover, other physical properties of lakes, such as freezing temperature, thermal conductivity, and saturated vapor pressure at the lake-air interface, are also



Earth and Space Science



Figure 9. Comparisons of seasonal lake surface water temperature between observations from ARC-Lake and simulations from the optimization experiment (Exp_optimization) for six subregions during 1996–2011. The black line is a linear fit line. The green line is the 1:1 line.

related to the salinity of the lake (Jackett et al., 2006; Sun et al., 2008; Wen et al., 2015). These processes are not described in FLake and will introduce some errors to the simulations of saline lakes in the TP region.

Additionally, we used the ARC-Lake satellite observations to evaluate and calibrate FLake. Generally, satellite-derived LSWT is slightly lower than buoy observations. The satellite product retrieves the surface skin temperature, while the buoy is placed in the bulk layer deeper in the water, which represents the water temperature of this layer. The discrepancy between skin and bulk temperature is known as the cool skin effect (Donlon et al., 2002; Minnett, 2003; Robinson et al., 1984). The widely positive biases between simulated





Figure 10. Taylor diagram of monthly lake surface water temperature of subregions in the open-water period for five experiments.



Figure 11. (a) Comparison of lake mean depth and modeled effective lake mean depth. (b) Comparison of the light extinction coefficient (K_d) and modeled effective light extinction coefficient. Color circles represent the effective depth (K_d) factors. The insert is the box plot of modeled effective lake mean depth (K_d) factors at different lake mean depths (K_d s).



LSWT and ARC-Lake observations across China during the open-water period may be partly attributed to this phenomenon. Future studies should utilize more in situ observations to further confirm the robustness of lake temperature simulations.

Data Availability Statement

Satellite-derived LSWTs used in the study are available to download from http://www.laketemp.net/home_ ARCLake/data_access.php (ARC-Lake, version 3.0). The geomorphological characteristics of the lakes, including latitude, longitude, surface area, and mean depth, used in this study were obtained from the Hydro-LAKES database (version 1.0, https://www.hydrosheds.org/pages/hydrolakes). The MODIS albedo product MCD34B3 (version 5) is available from the Land Processes Distributed Active Archive Center (LP DAAC; https://lpdaac.usgs.gov/data_access). Water clarity data in China are provided by Wang et al. (2020): Changes of water clarity in large lakes and reservoirs across China observed from long-term MODIS (https://doi. org/10.1016/j.rse.2020.111949). The China Meteorological Forcing Data set (CMFD) can be obtained from http://data.tpdc.ac.cn/en/data/8028b944-daaa-4511-8769-965612652c49/. The source code for FLake model is available at http://www.flake.igb-berlin.de/site/download. All the links were last accessed on June 27, 2021.

References

- Balsamo, G., Salgado, R., Dutra, E., Boussetta, S., Stockdale, T., & Potes, M. (2012). On the contribution of lakes in predicting near-surface temperature in a global weather forecasting model. *Tellus A: Dynamic Meteorology and Oceanography*, 64(1), 15829. https://doi. org/10.3402/tellusa.v64i0.15829
- Bastviken, D., Cole, J., Pace, M., & Tranvik, L. (2004). Methane emissions from lakes: Dependence of lake characteristics, two regional assessments, and a global estimate. *Global Biogeochemical Cycles*, 18(4). https://doi.org/10.1029/2004gb002238
- Bernhardt, J., Engelhardt, C., Kirillin, G., & Matschullat, J. (2012). Lake ice phenology in Berlin-Brandenburg from 1947–2007: Observations and model hindcasts. *Climatic Change*, 112(3), 791–817. https://doi.org/10.1007/s10584-011-0248-9
 - Butcher, J. B., Nover, D., Johnson, T. E., & Clark, C. M. (2015). Sensitivity of lake thermal and mixing dynamics to climate change. Climatic Change, 129(1), 295–305. https://doi.org/10.1007/s10584-015-1326-1
 - Cai, Y., Ke, C. Q., Li, X., Zhang, G., Duan, Z., & Lee, H. (2019). Variations of lake ice phenology on the Tibetan Plateau from 2001 to 2017 based on MODIS data. *Journal of Geophysical Research: Atmospheres*, 124(2), 825–843. https://doi.org/10.1029/2018jd028993
 - Carter, J. L., & Schindler, D. E. (2012). Responses of zooplankton populations to four decades of climate warming in lakes of southwestern Alaska. *Ecosystems*, 15(6), 1010–1026. https://doi.org/10.1007/s10021-012-9560-0
 - Chen, J. L., Pekker, T., Wilson, C. R., Tapley, B. D., Kostianoy, A. G., Cretaux, J. F., & Safarov, E. S. (2017). Long-term Caspian Sea level change. *Geophysical Research Letters*, 44(13), 6993–7001. https://doi.org/10.1002/2017gl073958
 - Donlon, C. J., Minnett, P. J., Gentemann, C., Nightingale, T. J., Barton, I. J., Ward, B., & Murray, M. J. (2002). Toward improved validation of satellite sea surface skin temperature measurements for climate research. *Journal of Climate*, 15(4), 353–369. https://doi. org/10.1175/1520-0442(2002)015<0353:tivoss>2.0.co;2
 - Feng, L., Hou, X., & Zheng, Y. (2019). Monitoring and understanding the water transparency changes of fifty large lakes on the Yangtze Plain based on long-term MODIS observations. *Remote Sensing of Environment*, 221, 675–686. https://doi.org/10.1016/j.rse.2018.12.007
- Fenocchi, A., Rogora, M., Sibilla, S., Ciampittiello, M., & Dresti, C. (2018). Forecasting the evolution in the mixing regime of a deep subalpine lake under climate change scenarios through numerical modelling (Lake Maggiore, Northern Italy/Southern Switzerland). *Climate Dynamics*, 51(9), 3521–3536. https://doi.org/10.1007/s00382-018-4094-6
- Gleick, P. H. (1993). Water and conflict: Fresh water resources and international security. *International Security*, 18(1), 79–112. https://doi.org/10.2307/2539033
- Goyette, S., McFarlane, N. A., & Flato, G. M. (2000). Application of the Canadian regional climate model to the Laurentian Great Lakes region: Implementation of a lake model. *Atmosphere-Ocean*, *38*(3), 481–503. https://doi.org/10.1080/07055900.2000.9649657
- Gronewold, A. D., & Stow, C. A. (2014). Water loss from the Great Lakes. Science, 343(6175), 1084–1085. https://doi.org/10.1126/ science.1249978
- He, J., Yang, K., Tang, W. J., Lu, H., Qin, J., Chen, Y., & Li, X. (2020). The first high-resolution meteorological forcing dataset for land process studies over China. *Scientific Data*, 7(1), 1–11. https://doi.org/10.1038/s41597-020-0369-y
- He, T., Liang, S. L., & Song, D. X. (2014). Analysis of global land surface albedo climatology and spatial-temporal variation during 1981–2010 from multiple satellite products. *Journal of Geophysical Research: Atmospheres*, 119(17), 10–281. https://doi.org/10.1002/2014jd021667
- Hostetler, S. W., Bates, G. T., & Giorgi, F. (1993). Interactive coupling of a lake thermal-model with a regional climate model. *Journal of Geophysical Research*, 98(D3), 5045–5057. https://doi.org/10.1029/92jd02843
 Huy, S. A. (1989). *Coastel metacrology*. Academic Press Inc.
- Hsu, S. A. (1988). Coastal meteorology. Academic Press Inc.
- Huang, A., Wang, Y., Dai, Y., Yang, K., Wei, N., Wen, L., et al. (2019). Evaluating and improving the performance of three 1-D lake models in a large deep lake of the central Tibetan Plateau. *Journal of Geophysical Research: Atmospheres*, 124(6), 3143–3167. https://doi.org/10.1029/2018JD029610
- Huang, A., Zhao, Y., Zhou, Y., Yang, B., Zhang, L. J., Dong, X. N., et al. (2016). Evaluation of multisatellite precipitation products by use of ground-based data over China. Journal of Geophysical Research: Atmospheres, 121(18), 10–654. https://doi.org/10.1002/2016jd025456
- Huang, L., Wang, J. B., Zhu, L. P., Ju, J. T., & Daut, G. (2017). The warming of large lakes on the Tibetan Plateau: Evidence from a lake model simulation of Nam Co, China, during 1979–2012. *Journal of Geophysical Research: Atmospheres*, 122(24), 13095–13107. https://doi.org/10.1002/2017jd027379

Acknowledgments

This study is supported by the Second Tibetan Plateau Scientific Expedition and Research Program (Grant No. 2019QZKK0405) and the NSFC project Basic Science Center for the Tibetan Plateau Earth System (BCTES, Grant No. 41988101). The authors thank the two anonymous reviewers and the Editor C. Gentemann for their insightful comments.



Jackett, D. R., McDougall, T. J., Feistel, R., Wright, D. G., & Griffies, S. M. (2006). Algorithms for density, potential temperature, conservative temperature, and the freezing temperature of seawater. *Journal of Atmospheric and Oceanic Technology*, 23(12), 1709–1728. https:// doi.org/10.1175/Jtech1946.1

Jeppesen, E., Meerhoff, M., Davidson, T. A., Trolle, D., Sondergaard, M., Lauridsen, T. L., et al. (2014). Climate change impacts on lakes: An integrated ecological perspective based on a multi-faceted approach, with special focus on shallow lakes. *Journal of Limnology*, 73(s1), 88–111. https://doi.org/10.4081/jlimnol.2014.844

- Lang, J., Lyu, S., Li, Z., Ma, Y., & Su, D. (2018). An investigation of ice surface albedo and its influence on the high-altitude lakes of the Tibetan Plateau. *Remote Sensing*, 10(2), 218. https://doi.org/10.3390/rs10020218
- Layden, A., MacCallum, S. N., & Merchant, C. J. (2016). Determining lake surface water temperatures worldwide using a tuned one-dimensional lake model (FLake, v1). Geoscientific Model Development, 9(6), 2167–2189. https://doi.org/10.5194/gmd-9-2167-2016
- Layden, A., Merchant, C., & MacCallum, S. (2015). Global climatology of surface water temperatures of large lakes by remote sensing. International Journal of Climatology, 35(15), 4464–4479. https://doi.org/10.1002/joc.4299
- Lazhu Yang, K., Wang, J., Lei, Y., Chen, Y., Ding, B., Zhu, L., et al. (2016). Quantifying evaporation and its decadal change for Lake Nam Co, central Tibetan Plateau. *Journal of Geophysical Research: Atmospheres*, 121(13), 7578–7591. https://doi.org/10.1002/2015jd024523
- Lei, Y., Zhu, Y., Wang, B., Yao, T., Yang, K., Zhang, X., et al. (2019). Extreme lake level changes on the Tibetan Plateau associated with the 2015/2016 El Niño. Geophysical Research Letters, 46(11), 5889–5898. https://doi.org/10.1029/2019gl081946
- Le Moigne, P., Colin, J., & Decharme, B. (2016). Impact of lake surface temperatures simulated by the FLake scheme in the CNRM-CM5 climate model. *Tellus A: Dynamic Meteorology and Oceanography*, 68(1), 31274. https://doi.org/10.3402/tellusa.v68.31274
- Li, X., Peng, S., Deng, X., Su, M., & Zeng, H. (2019). Attribution of lake warming in four shallow lakes in the middle and lower Yangtze River Basin. Environmental Science & Technology, 53(21), 12548–12555. https://doi.org/10.1021/acs.est.9b03098
- Li, Y., Wang, T., Zeng, Z., Peng, S., Lian, X., & Piao, S. (2016). Evaluating biases in simulated land surface albedo from CMIP5 global climate models. Journal of Geophysical Research: Atmospheres, 121(11), 6178–6190. https://doi.org/10.1002/2016jd024774
- Li, Z., Ao, Y., Lyu, S., Lang, J., Wen, L., Stepanenko, V., et al. (2018). Investigation of the ice surface albedo in the Tibetan Plateau lakes based on the field observation and MODIS products. *Journal of Glaciology*, *64*(245), 506–516. https://doi.org/10.1017/jog.2018.35
- Lv, Z., Zhang, S., Jin, J., Wu, Y., & Ek, M. B. (2019). Coupling of a physically based lake model into the climate forecast system to improve winter climate forecasts for the Great Lakes region. *Climate Dynamics*, 53(9), 6503–6517. https://doi.org/10.1007/s00382-019-04939-2
 Ma, B., Yang, C., Duan, H., Jiang, L., Wang, S., Fang, Y., et al. (2011). *Chinale lakes at present: Number area and partial distribution.* Sciences 2013.
- Ma, R., Yang, G., Duan, H., Jiang, J., Wang, S., Feng, X., et al. (2011). China's lakes at present: Number, area and spatial distribution. Science China Earth Sciences, 54(2), 283–289. https://doi.org/10.1007/s11430-010-4052-6
- MacCallum, S. N., & Merchant, C. J. (2012). Surface water temperature observations of large lakes by optimal estimation. Canadian Journal of Remote Sensing, 38(1), 25–45. https://doi.org/10.5589/m12-010
- Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geo-statistical approach. *Nature Communications*, 7, 13603. https://doi.org/10.1038/ncomms13603
- Minnett, P. (2003). Radiometric measurements of the sea-surface skin temperature: The competing roles of the diurnal thermocline and the cool skin. *International Journal of Remote Sensing*, 24(24), 5033–5047. https://doi.org/10.1080/0143116031000095880
- Mironov, D. (2008). Parameterization of lakes in numerical weather prediction. Description of a lake model (COSMO Technical Report No. 11). Offenbach am Main, Germany: Deutscher Wetterdienst.
- Mironov, D., Heise, E., Kourzeneva, E., Ritter, B., Schneider, N., & Terzhevik, A. (2010). Implementation of the lake parameterisation scheme FLake into the numerical weather prediction model COSMO. *Boreal Environment Research*, 15(2), 218–230.
- O'Reilly, C. M., Alin, S. R., Plisnier, P. D., Cohen, A. S., & McKee, B. A. (2003). Climate change decreases aquatic ecosystem productivity of Lake Tanganyika, Africa. Nature, 424(6950), 766–768. https://doi.org/10.1038/nature01833
- O'Reilly, C. M., Sharma, S., Gray, D. K., Hampton, S. E., Read, J. S., Rowley, R. J., et al. (2015). Rapid and highly variable warming of lake surface waters around the globe. *Geophysical Research Letters*, 42(24), 10773–10781. https://doi.org/10.1002/2015gl066235
- Paerl, H. W., & Paul, V. J. (2012). Climate change: Links to global expansion of harmful cyanobacteria. *Water Research*, 46(5), 1349–1363. https://doi.org/10.1016/j.watres.2011.08.002
- Perroud, M., Goyette, S., Martynov, A., Beniston, M., & Anneville, O. (2009). Simulation of multiannual thermal profiles in deep Lake Geneva: A comparison of one-dimensional lake models. *Limnology & Oceanography*, 54(5), 1574–1594. https://doi.org/10.4319/ lo.2009.54.5.1574
- Phillips, W., & Irbe, J. (1978). Land-to-lake comparison of wind, temperature, and humidity on lake Ontario during the international field Year for the Great lakes (IFYGL) rep, CLI-2-77. Atmospheric Environment Service.
- Pour, H. K., Duguay, C. R., Martynov, A., & Brown, L. C. (2012). Simulation of surface temperature and ice cover of large northern lakes with 1-D models: A comparison with MODIS satellite data and in situ measurements. *Tellus A: Dynamic Meteorology and Oceanography*, 64(1), 17614. https://doi.org/10.3402/tellusa.v64i0.17614
- Quinn, F. H. (1979). An improved aerodynamic evaporation technique for large lakes with application to the International Field Year for the Great Lakes. Water Resources Research, 15(4), 935–940. https://doi.org/10.1029/wr015i004p00935
- Robinson, I. S., Wells, N. C., & Charnock, H. (1984). The sea-surface thermal-boundary layer and its relevance to the measurement of sea-surface temperature by airborne and spaceborne radiometers. *International Journal of Remote Sensing*, 5(1), 19–45. https://doi. org/10.1080/01431168408948787
- Rooney, G. G., & Bornemann, F. J. (2013). The performance of FLake in the met office unified model. *Tellus A: Dynamic Meteorology and Oceanography*, 65(1), 21363. https://doi.org/10.3402/tellusa.v65i0.21363
- Schaaf, C. B., Gao, F., Strahler, A. H., Lucht, W., Li, X., Tsang, T., et al. (2002). First operational BRDF, albedo nadir reflectance products from MODIS. *Remote Sensing of Environment*, 83(1–2), 135–148. https://doi.org/10.1016/s0034-4257(02)00091-3
- Sharma, S., Blagrave, K., Magnuson, J. J., O'Reilly, C. M., Oliver, S., Batt, R. D., et al. (2019). Widespread loss of lake ice around the Northern Hemisphere in a warming world. *Nature Climate Change*, 9(3), 227–231. https://doi.org/10.1038/s41558-018-0393-5
- Song, C. Q., Huang, B., & Ke, L. H. (2013). Modeling and analysis of lake water storage changes on the Tibetan Plateau using multi-mission satellite data. *Remote Sensing of Environment*, 135, 25–35. https://doi.org/10.1016/j.rse.2013.03.013
- Steiner, A. L., Posselt, D. J., & Wright, D. M. (2013). Sensitivity of lake-effect snowfall to lake ice cover and temperature in the Great Lakes region. Monthly Weather Review, 141(2), 670–689. https://doi.org/10.1175/mwr-d-12-00038.1
- Stepanenko, V. M., Goyette, S., Martynov, A., Perroud, M., Fang, X., & Mironov, D. (2010). First steps of a lake model intercomparison project: LakeMIP. Boreal Environment Research, 15(2), 191–202.
- Stepanenko, V. M., Martynov, A., Jöhnk, K. D., Subin, Z. M., Perroud, M., Fang, X., et al. (2013). A one-dimensional model intercomparison study of thermal regime of a shallow, turbid midlatitude lake. *Geoscientific Model Development*, 6(4), 1337–1352. http://doi.org/10.5194/gmd-6-1337-2013

- Su, D., Hu, X., Wen, L., Lyu, S., Gao, X., Zhao, L., et al. (2019). Numerical study on the response of the largest lake in China to climate change. *Hydrology and Earth System Sciences*, 23(4), 2093–2109. https://doi.org/10.5194/hess-23-2093-2019
- Subin, Z. M., Riley, W. J., & Mironov, D. (2012). An improved lake model for climate simulations: Model structure, evaluation, and sensitivity analyses in CESM1. Journal of Advances in Modeling Earth Systems, 4. https://doi.org/10.1029/2011ms000072
- Sun, H. B., Feistel, R., Koch, M., & Markoe, A. (2008). New equations for density, entropy, heat capacity, and potential temperature of a saline thermal fluid. Deep Sea Research Part I: Oceanographic Research Papers, 55(10), 1304–1310. https://doi.org/10.1016/j.dsr.2008.05.011
- Taylor, K. E. (2001). Summarizing multiple aspects of model performance in a single diagram. Journal of Geophysical Research: Atmospheres, 106(D7), 7183–7192. https://doi.org/10.1029/2000jd900719
- Thiery, W., Davin, E. L., Panitz, H. J., Demuzere, M., Lhermitte, S., & Van Lipzig, N. (2015). The Impact of the African Great Lakes on the regional climate. *Journal of Climate*, *28*(10), 4061–4085. https://doi.org/10.1175/jcli-d-14-00565.1
- Turuncoglu, U. U., Elguindi, N., Giorgi, F., Fournier, N., & Giuliani, G. (2013). Development and validation of a regional coupled atmosphere lake model for the Caspian Sea Basin. *Climate Dynamics*, 41(7), 1731–1748. https://doi.org/10.1007/s00382-012-1623-6
- Verburg, P., & Hecky, R. E. (2009). The physics of the warming of Lake Tanganyika by climate change. *Limnology & Oceanography*, 54(6), 2418–2430. http://doi.org/10.4319/lo.2009.54.6_part_2.2418
- Verseghy, D. L., & MacKay, M. D. (2017). Offline Implementation and evaluation of the Canadian small lake model with the Canadian land surface scheme over Western Canada. *Journal of Hydrometeorology*, 18(6), 1563–1582. https://doi.org/10.1175/Jhm-D-16-0272.1
- Wang, S., Li, J., Zhang, B., Lee, Z., Spyrakos, E., Feng, L., et al. (2020). Changes of water clarity in large lakes and reservoirs across China observed from long-term MODIS. *Remote Sensing of Environment*, 247, 111949. https://doi.org/10.1016/j.rse.2020.111949
- Wen, L. J., Nagabhatla, N., Zhao, L., Li, Z. G., & Chen, S. Q. (2015). Impacts of salinity parameterizations on temperature simulation over and in a hypersaline lake. *Chinese Journal of Oceanology and Limnology*, 33(3), 790–801. https://doi.org/10.1007/s00343-015-4153-3
- Woolway, R. I., Kraemer, B. M., Lenters, J. D., Merchant, C. J., O'Reilly, C. M., & Sharma, S. (2020). Global lake responses to climate change. *Nature Reviews Earth & Environment*, 1(8), 388–403. https://doi.org/10.1038/s43017-020-0067-5
- Woolway, R. I., Meinson, P., Nöges, P., Jones, I. D., & Laas, A. (2017). Atmospheric stilling leads to prolonged thermal stratification in a large shallow polymictic lake. *Climatic Change*, 141(4), 759–773. https://doi.org/10.1007/s10584-017-1909-0
- Woolway, R. I., & Merchant, C. J. (2019). Worldwide alteration of lake mixing regimes in response to climate change. *Nature Geoscience*, 12(4), 271–276. http://doi.org/10.1038/s41561-019-0322-x
- Wu, Y., Huang, A., Yang, B., Dong, G., Wen, L., Zhang, Z., et al. (2019). Numerical study on the climatic effect of the lake clusters over Tibetan Plateau in summer. *Climate Dynamics*, 53(9), 5215–5236. https://doi.org/10.1007/s00382-019-04856-4
- Wu, Y., Huang, A., Yang, X., Qiu, B., Wen, L., Wen, L., et al. (2020). Improvements of the coupled WRF-Lake model over Lake Nam Co, central Tibetan Plateau. *Climate Dynamics*, 55, 2703–2724. https://doi.org/10.1007/s00382-020-05402-3
- Xu, X. B., Jiang, B., Tan, Y., Costanza, R., & Yang, G. S. (2018). Lake-wetland ecosystem services modeling and valuation: Progress, gaps and future directions. *Ecosystem Services*, 33, 19–28. https://doi.org/10.1016/j.ecoser.2018.08.001
- Yang, K., He, J., Tang, W. J., Qin, J., & Cheng, C. C. K. (2010). On downward shortwave and longwave radiations over high altitude regions: Observation and modeling in the Tibetan Plateau. Agricultural and Forest Meteorology, 150(1), 38–46. https://doi.org/10.1016/j. agrformet.2009.08.004
- Yang, X., & Lu, X. (2014). Drastic change in China's lakes and reservoirs over the past decades. *Scientific Reports*, 4, 6041. https://doi. org/10.1038/srep06041
- Zhang, G., Yao, T., Chen, W., Zheng, G., Shum, C. K., Yang, K., et al. (2019). Regional differences of lake evolution across China during 1960s–2015 and its natural and anthropogenic causes. *Remote Sensing of Environment*, 221, 386–404. https://doi.org/10.1016/j. rse.2018.11.038
- Zhang, G., Yao, T., Piao, S., Bolch, T., Xie, H., Chen, D., et al. (2017). Extensive and drastically different alpine lake changes on Asia's high plateaus during the past four decades. *Geophysical Research Letters*, 44(1), 252–260. https://doi.org/10.1002/2016gl072033
- Zhang, G., Yao, T., Xie, H., Qin, J., Ye, Q., Dai, Y., & Guo, R. (2014). Estimating surface temperature changes of lakes in the Tibetan Plateau using MODIS LST data. *Journal of Geophysical Research: Atmospheres*, 119(14), 8552–8567. https://doi.org/10.1002/2014jd021615